Author's response to reviews

Title: Stage I seminoma: Treatment outcomes in a low-income country

Authors:

Jamal Khader (jkhader@khcc.org)
Ahmed Salem (alwikah@hotmail.com)
Yazan Abuodeh (yabuodeh@khcc.org)
Abdelatief Almousa (aalmousa@khcc.org)
Naim Farah (nfarah@khcc.org)
Fadwa Abdelrahman (fdeeb@khcc.org)

Version: 2 Date: 16 December 2011

Author's response to reviews:

Dear Editor,

Please find the attached revised manuscript entitled "Stage I seminoma: Treatment outcomes in a low-income country" intended for publication in the BMC Urology.

We have added data relating to income and insurance status.

Please note that the authors have not submitted this paper to any other journal.

Thank you for considering this manuscript for publication.

King regards,

Jamal Khader MD, Ahmed Salem MD, Yazan Abuodeh MD, Abdelateif Almousa MD, PhD, Naim Farah MD, Fadwa Abdelrahman MD

King Hussein Cancer Center

P.O.Box: 142 730, Amman 11814

Amman, Jordan

Tel.: +962 788 305344

Fax: +962 658 57323

Email: alwikah@hotmail.com